Author's response to reviews

**Title:** Cost-effectiveness of recurrence risk guided care versus care as usual in women who suffered from early-onset preeclampsia including HELLP syndrome in their previous pregnancy (the PreCare study)

**Authors:**

Denise HJ Delahaije (denise.delahaije@mumc.nl)
Sander MJ van Kuijk (sander.vankuijk@epid.unimaas.nl)
Carmen D Dirksen (c.dirksen@mumc.nl)
Simone JS Sep (sjs.sep@epid.unimaas.nl)
Louis L Peeters (l.peeters@og.unimaas.nl)
Marc E Spaanderman (m.spaanderman@obgyn.umcn.nl)
Hein W Bruinse (h.w.bruinse@umcutrecht.nl)
Laura D de Wit (lzuurendonk@hotmail.com)
Joris AM van der Post (j.a.vanderpost@amc.uva.nl)
Johannes J Duvekot (j.j.duvekot@erasmusmc.nl)
Jim van Eyck (j.van.eyck@isala.nl)
Mariëlle G van Pampus (m.g.van.pampus@og.umcg.nl)
Mark ABHM van der Hoeven (m.vander.hoeven@mumc.nl)
Luc J Smits (luc.smits@epid.unimaas.nl)

**Version:** 2  **Date:** 11 May 2010

**Author's response to reviews:**

D.H.J. Delahaije, MSc
Maastricht University Medical Centre
Clinical Epidemiology and MTA
P.O. Box 5800
6202 AZ Maastricht
The Netherlands

Maastricht, 3 May 2010

Dear Sir/Madam,

We are pleased to submit our manuscript entitled ‘Cost-effectiveness of recurrence risk guided care versus care as usual in women who suffered from early-onset preeclampsia including HELLP syndrome in their previous pregnancy (the PreCare study)’ for publication in BMC Pregnancy and Childbirth.

This article contains a study protocol of an ongoing study. We believe this manuscript is of interest to the readers of BMC Pregnancy and Childbirth, as it provides a clear description including a detailed account of the rationale and methodology of the study.
We hope this paper meets the requirements for publication in BMC Pregnancy and Childbirth.

Yours sincerely, on behalf of the co-authors,

Denise Delahaije